Author's response to reviews

Title: Volume-based referral for cardiovascular procedures in the United States: a cross-sectional regression analysis

Authors:

Andrew J Epstein (andrew.epstein@yale.edu)
Saif S Rathore (saif.rathore@yale.edu)
Kevin G M Volpp (volpp70@mail.med.upenn.edu)
Harlan M Krumholz (harlan.krumholz@yale.edu)

Version: 3 Date: 27 May 2005

Author's response to reviews:

All revisions involved minor copyediting.
Per the Editor's request, we:
1) changed the title to sentence case,
2) removed authors' qualifications,
3) placed references before any full stops, and
4) changed the formatting for Table 1 so that it spans two pages and does not include a second title or set of column headers.

Also, we fixed some minor errors in the References section, including removing a duplicate cite and updating the information for the internet sites.

Thank you.

Andrew J. Epstein